Author's response to reviews

Title: Relationship between knee and ankle degeneration in a population of organ donors.

Authors:

Carol Muehleman (carol_muehleman@rush.edu)
Arkady margulis (arkady_margulis@rush.edu)
Won Bae (wbae@ucsd.edu)
Koichi Masuda (kmasuda@ucsd.edu)

Version: 3 Date: 7 July 2010

Author's response to reviews: see over
July 7, 2010

Melissa Norton, M.D.
Editor-in-Chief
BMC Medicine

Manuscript: #1409914228416459

Dear Dr. Norton,

My coauthors and I respectfully resubmit the original manuscript entitled “Relationship Between Knee and Ankle Degeneration in a Population of Organ Donors” with the editorial revisions requested. This article was originally submitted to BMC Musculoskeletal and was forwarded by the editor to you for consideration for publication.

The changes requested by the editorial staff include the correction to the second author of the manuscript and the delineation of the Conclusion section. These have been carried out.

We are delighted that you and your editorial staff deem our work suitable for your excellent journal.

Sincerely,

Carol Muehleman, Ph.D.
Professor
Department of Biochemistry
Department of Anatomy and Cell Biology
Department of Orthopedic Surgery